\xname{H2-Air12}
\W\subsubsection{Minimum tube diameter vs equivalence ratio; H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air12.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air12.gif}}
\T\caption{Minimum tube diameter vs equivalence ratio; H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air12.ps}

\T at26 - Table~\ref{at26}~\cite[Agafonov (1994)]{g_agaf94}
\W\xlink{at26.txt}{\rawdatadir/at26.txt}
\W Agafonov (1994) \cite{g_agaf94}
T=298 K, P=100 kPa

\T\clearpage

\xname{H2-Ox15}
\W\subsubsection{Minimum tube diameter vs equivalence ratio; H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox15.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox15.gif}}
\T\caption{Minimum tube diameter vs equivalence ratio; H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox15.ps}

\T at27a - Table~\ref{at27a}~\cite[Agafonov (1994)]{g_agaf94}
\W\xlink{at27a.txt}{\rawdatadir/at27a.txt}
\W Agafonov (1994) \cite{g_agaf94}
T=298 K, P=100 kPa

\T at27b - Table~\ref{at27b}~\cite[Agafonov (1994)]{g_agaf94}
\W\xlink{at27b.txt}{\rawdatadir/at27b.txt}
\W Agafonov (1994) \cite{g_agaf94}
T=135 K, P=100 kPa

\T\clearpage

\xname{H2-Ox16}
\W\subsubsection{Minimum tube diameter vs percent diluent; H2-O2-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox16.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox16.gif}}
\T\caption{Minimum tube diameter vs percent diluent; H2-O2-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox16.ps}

\T at28a - Table~\ref{at28a}~\cite[Agafonov (1994)]{g_agaf94}
\W\xlink{at28a.txt}{\rawdatadir/at28a.txt}
\W Agafonov (1994) \cite{g_agaf94}
T=298 K, P=100 kPa, ER=1, 80-90\% Ar

\T at28b - Table~\ref{at28b}~\cite[Agafonov (1994)]{g_agaf94}
\W\xlink{at28b.txt}{\rawdatadir/at28b.txt}
\W Agafonov (1994) \cite{g_agaf94}
T=298 K, P=100 kPa, ER=1, 70-80\% He

\T at28c - Table~\ref{at28c}~\cite[Agafonov (1994)]{g_agaf94}
\W\xlink{at28c.txt}{\rawdatadir/at28c.txt}
\W Agafonov (1994) \cite{g_agaf94}
T=298 K, P=100 kPa, ER=1, 55-75\% N2

\T at28d - Table~\ref{at28d}~\cite[Agafonov (1994)]{g_agaf94}
\W\xlink{at28d.txt}{\rawdatadir/at28d.txt}
\W Agafonov (1994) \cite{g_agaf94}
T=298 K, P=100 kPa, ER=1, 90\% Ar

\T at28e - Table~\ref{at28e}~\cite[Agafonov (1994)]{g_agaf94}
\W\xlink{at28e.txt}{\rawdatadir/at28e.txt}
\W Agafonov (1994) \cite{g_agaf94}
T=298 K, P=100 kPa, ER=1, 86\% He

\T\clearpage

\xname{CH4-Ox11}
\W\subsubsection{Minimum tube diameter vs equivalence ratio; CH4-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox11.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox11.gif}}
\T\caption{Minimum tube diameter vs equivalence ratio; CH4-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox11.ps}

\T at133 - Table~\ref{at133}~\cite[Pusch (1962)]{w_pusc62}
\W\xlink{at133.txt}{\rawdatadir/at133.txt}
\W Pusch (1962) \cite{w_pusc62}
T=293 K, P=101.3 kPa

\T\clearpage

